Author’s response to reviews

Title: EGFR modulates complement activation in head and neck squamous cell carcinoma

Authors:

Anas Haider Abu-Humaidan (a.abuhumaidan@ju.edu.jo)
Lars Ekblad (lars.ekblad@med.lu.se)
Johan Wennerberg (johan.wennerberg@med.lu.se)
Ole Sørensen (ole_e.sorensen@med.lu.se)

Version: 2 Date: 05 Feb 2020

Author’s response to reviews:

Editor Comments:

1. In the section 'Funding', please also describe the role of the funding body/bodies in the design of the study and collection, analysis, and interpretation of data and in writing the manuscript.

Done.

2. Please add a “Conclusions” section after the “Discussion” section. This should state clearly the main conclusions of the research article and give a clear explanation of their importance and relevance.

Done.

3. If abbreviations are used in the text they should be defined in the text at first use, and a list of abbreviations should be provided in the Declarations.

Done.

4. Your manuscript needs to be copyedited. We recommend that you ask a native English speaking colleague to help you copyedit the paper. If this is not possible, you may need to use a professional language editing service. For authors who wish to have the language in their manuscript edited by a native-English speaker with scientific expertise, BioMed Central has a new in house editing service. The new editing tool can provide both scientific and language editing: http://authorservices.springernature.com/
The manuscript was copyedited by a native English-speaking colleague.

5. At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files. Please ensure that all figures, tables and additional/supplementary files are cited within the text.

Done.